



**CESIS** Electronic Working Paper Series

Paper No. 283

# How can firms benefit from access to knowledge intense services?

Hans Lööf Pardis Nabavi Börje Johansson

August, 2014

The Royal Institute of technology Centre of Excellence for Science and Innovation Studies (CESIS) http://www.cesis.se

# How can firms benefit from access to knowledge intense services?

Hans Lööf<sup>\*</sup>, Pardis Nabavi<sup>‡</sup>and Börje Johansson<sup>§</sup>

June 27, 2014

#### Abstract

This paper examines variation in a productivity growth within a given location and between different locations. Implementing a dynamic panel data approach on Swedish micro data, we test the separate and complementary effect of innovation and spillovers from the local milieu. Measuring potential knowledge spillover by access to knowledge intensive services, the estimation results produce strong evidence of differences in the capacity to benefit from external knowledge among persistent innovators, temporary innovators and non-innovators. The results are consistent regardless of whether innovation efforts are measured in terms of the frequency of patent applications or R&D investments. We do not find any differences in growth among non-innovative firms across locations, while the growth rate increases with the access to external knowledge for innovative firms and especially for persistent innovators.

Keywords: Innovation, spillovers, productivity, TFP-growth JEL-Codex: C23, O31, O32

<sup>†</sup>Corresponding author

<sup>\*</sup>Centre of Excellence for Science and Innovation Studies, Royal Institute of Technology Stockholm

 $<sup>^{\</sup>ddagger}\mathrm{Centre}$  of Excellence for Science and Innovation Studies, Royal Institute of Technology Stockholm

<sup>&</sup>lt;sup>§</sup>Centre of Excellence for Science and Innovation Studies, Royal Institute of Technology, Jönköping International Business School, CIRCLE at Lund University

# **1** INTRODUCTION

Given the attention that firms' engagement in research and innovation and their geographical location have attracted in the last decades, surprisingly few studies have been able to assess the separate and complementary effect of these two production factors within the same framework.

In order to fill this gap, our paper provides empirical evidence on the impact of internal knowledge generation and knowledge spillovers from neighbouring firms on firms' productivity growth. We proxy the first factor by the frequency of innovation efforts, and the second by the intensity of local knowledge sources.

In a dynamic framework, we then consider various innovation strategies in a particular location, and a particular innovation strategy in different locations. Using this methodology, we are able to test if firms with persistent frequency of innovation activities can overcome a low level of external knowledge potential, and whether a high external knowledge potential can compensate for a low level of internal knowledge.

Our empirical analysis applies to Swedish firm-level observations on manufacturing and service firm and we study two different time periods. In cases where patent applications are used to capture the innovation strategy (or internal knowledge generation) of the firm, the period studied extends from 1997 to 2008. In the second case, where R&D engagement information is used, the study employs survey data from three consecutive Swedish Community Innovation Surveys (CIS), covering the period 2002-2008. We use access to knowledgeintensive producer services as indicator for the mass or amount of influential external knowledge in the local milieu. In the empirical analysis, the paper identifies 35 Swedish producer-service industries at the 5-digit level in which more than 30 per cent of employees have a university degree. These services include ICT services, engineering R&D and engineering services, financial services, and the brokerage and recruitment of manpower.

The estimation results produce strong evidence of differences in the capacity to benefit from external knowledge among persistent innovators, temporary innovators and non-innovators. The results are consistent regardless of whether innovation efforts are measured in terms of the frequency of patent applications or R&D investments. We do not find any differences in growth among non-innovative firms across locations, while the growth rate increases with the access to external knowledge for innovative firms.

The remainder of the paper is organised as follows: The next section discusses the relevant literature on internal and external knowledge and clarifies our theoretical assumptions. It also presents the hypotheses to be tested. Section 3 describes the data, and Section 4 introduces the testing strategy and the associated model specifications. Section 5 discusses and interprets the main findings, and Section 6 concludes.

# 2 LITERATURE REVIEW AND HYPOTHESES

The objective of our research is to clarify the idea of a symbiotic relationship between research and innovation (R&I) and spillovers by distinguishing between different levels and combinations of internal and external knowledge. In this section we will motivate our hypotheses by first briefly review literature on heterogeneity and persistence concerning firms' productivity and innovation activities. We then consider literature on spatial proximity to knowledge with business potential. Finally, we discuss literature on knowledge-intense producer service, and our intention is to show that this category of enterprises play an important role in the process of knowledge spillovers. Based on the reviewed literature, we formulate the four hypotheses about innovation and the local environment to be tested on two different panels of firms in 290 different Swedish municipalities.

#### 2.1 Innovation, location and spillovers

A growing body of empirical literature documents the existence of performance heterogeneity across firms and establishments. This observation remains valid for several performance measures, including profitability, productivity and growth (Bartelsman and Doms, 2000). To a large extent, the heterogeneity also tends to persist over long periods (Mueller 1986, Cubbin and Geroski 1987, Geroski and Jacquemin 1988, Geroski 1998, Gschwandtner 2004, Syverson 2004 and 2011, Dosi 2007). Surveying the micro literature, Dosi and Nelson (2010) find that the heterogeneous productivity pattern can be explained by different abilities to innovate and/or adopt innovations developed elsewhere.

Previous studies have documented that in most fields of innovation and technology, progress is cumulative in the sense that today's efforts build on preceding efforts. Prior experience from related projects can create internal capabilities within organisations, and learning economies and these categories of internal spillovers tend to reduce the costs of new innovation (Nelson and Winter 1982; Attewell 1992; Cohen and Levinthal 1990; Åstebro 2002; Phene and Almeida, 2008, Teece 2010). Thus, the continuity of innovation efforts ensures the accumulation of internal knowledge, whereas disruption can cause the erosion and obsolescence of acquired skills, routines and technology.

The overall picture that emerges from recent empirical studies, however indicates that many firms are not at all engaged in innovation and R&D activities, some are innovation active only occasionally, and others remain persistently innovation active over periods of years (Cefis and Orsenigo, 2001, Duguet and Monjon 2002, Klette and Kortum 2004, Peters 2009, Peters et al. 2013).

The literature suggests various explanations for firms' selection into persistence of innovation or not. One element of the literature stems from evolutionary theory and emphasises the importance of technological trajectories. Along the technological trajectory, firms learn by innovating and developing organisational competencies (Raymond et al. 2010). Other explanations include the relationships between innovation and market power or financial constraints as selection mechanisms (Brown et al. 2009).

Firms do not learn solely from internal spillovers. A common element of many theoretical propositions in the productivity literature and related economic models (Marshall 1890, Arrow 1962, Jacobs 1969, Romer 1986, Porter 1990) is the hypothesis that a firm or an industry benefits from spatial proximity to knowledge. The presence of external knowledge flows should reveal itself in social returns to innovation efforts in addition to private returns<sup>1</sup>.

Similar to the heterogeneity in performance across firms, there is also heterogeneity across regions and firms concerning distribution and absorption of spillovers. Numerous studies have clarified that the social rate of return is larger in agglomeration areas, and that knowledge flows decline in volume and intensity as the distance between origin and destination grows.<sup>2</sup> <sup>3</sup>

A local environment with a wide spectrum of knowledge resources and a wide range of qualifications and competence profiles regarding the labour supply provides rich opportunities for knowledge exchange and creative interaction between firms and individuals. As a rule, these features apply to large urban regions in which the knowledge potential is higher than elsewhere. The importance of proximity between suppliers and buyers of knowledge-intensive producer services can be linked to the theory of agglomeration economies in large urban regions, according to which large regions offer companies more positive externalities than small regions. Fujita and Thisse (2002) describe this phenomenon as "communication externalities". They measure the extent of the agglomeration advantage of a single firm by the company's accessibility to other companies.

A growing number of studies (Arora and Gambardella 1994, Hagedoorn 2002, Cassiman and Veugelers 2002 and 2006, Escribano et al. 2009, to mention a few) emphasize the importance of in-house capacity for absorbing external knowledge, consistent with seminal papers by Cohen and Levinthal (1990) and

 $<sup>^{1}</sup>$ In a recent analysis based on technology flows across industries, Wolff (2012) finds that the direct rate of return to R&D in the US between 1958 and 2007 was 22%, whereas the indirect rate of return to R&D was 37%.

<sup>&</sup>lt;sup>2</sup>Friction costs vary for both non-market spillover and commercial transfers because of communication distances. Distance frictions increase when knowledge is complex (Beckmann 2000) and when it is tacit (Polanyi 1966). Knowledge also has a tendency to be spatially sticky (von Hippel 1994).

<sup>&</sup>lt;sup>3</sup>Consistent with predictions from gravity models, Glaeser and Gottlieb (2009) estimate that the elasticity of income with respect to city size in the U.S. is within the range of 0.04-0.08 for different model specifications. The size of the estimates is comparable with estimates by Ciccone and Hall (1996), who find that a doubling of employment density in a county results in a 6% increase in average labour productivity. Using patent data, several studies indicate that the number of cross-citations significantly decreases as distance increases (Maurseth and Verspagen 1999, Verspagen and Schoenmakers 2000). Testing hypotheses on variety, Frenken, van Oort, and Verburg (2007) report that Dutch regions with a high degree of related variety had the highest rates of growth in employment. Focusing explicitly on innovations, Brouwer et al. (1999) demonstrate that firms located in agglomerated Dutch regions tend to produce larger numbers of new products than firms located in more rural regions. Similar result is reported by Doloreux and Shearmur (2012).

Rosenberg (1990). This research question the widespread assumption in economic geography that is all firms in a milieu such as a cluster or an agglomeration may benefit from access to a high concentration of specialized, supplemented or varied knowledge diffused through voluntary and involuntary mechanisms. Further counterarguments are provided by recent research on ICT clusters in Cambridge (Huber 2012) and agglomerations in Sweden (Lööf and Johansson, 2012), which both argue that a large fraction of firms do not have any real benefit or only limited benefit from being located in knowledge-intensive places.

However, the general importance of access to external knowledge tend to increase in a knowledge-based innovation driven economy. In their survey of the literature, Breschi and Lissoni (2001) conclude that "when firms are constantly innovating and are frequently changing process and product configurations, there is the need to be close to a constellation of allied firms and specialised suppliers to smooth input-output linkages". Consistent with this reasoning, several work suggest that a focusation only on internal knowledge and the development of internal capabilities and routines is no longer sufficient for coping with challenges such as shorter product life cycles, greater technological complexity, more specialised knowledge and increasing cost. Therefore, firms need to tao into external knowledge sources. For early contribution within this strand of the literature, see Kogut and Zander (1992).

Producer services generally represent market-supporting services that improve the allocating efficiency of a the economy and thus enhance productivity of individual firms. Knowledge-intensive producer services expand the potential for firms to interact with a wide range of knowledge-intensive services, providing rich opportunities for knowledge exchange and creative interaction that eventually results in technological development based on process and/or product innovation. In both cases, the buyer of the services will benefit from agglomeration externalities because knowledge-intensive service firms seek to sell their services and specialised knowledge to more than one client company, indirectly transmitting novel concepts and solutions from one customer to another.

A non-negligible fraction of the knowledge flows across firms are related to links outside the nearby milieu of the firms. Several groups of previous studies address this issue, such as the literature on (i) multinational corporations (e.g., Almeida and Phene 2012, Nobel and Birkinshaw 1998), (ii) strategic alliances and R&D collaboration (Hagedoorn 2002, Belderbos et al. 2012), (iii) urban economics (Saxenian 1990, Almedia 1996) and (iv) cognitive proximity (Nooteboom et al. 2007, Lychagin et al. 2010). Although the question of knowledge flows from more distant locations is beyond the scope of our study, we account for remote knowledge diffusion to some extent by including foreign and domestic multinational corporate ownership among the controls.

#### 2.2 General approach and hypotheses

The general approach of this paper is the following: First, we group the observed Swedish firms into three categories reflecting their internal knowledge. Second, the external knowledge potential of each firm is also arranged into three categories. These two steps allow us to classify the firms into nine different categories.

With regard to the internal knowledge, three classification are defined. The first includes firms that do not engage in research and innovation activities (i.e., patent applications in one of the samples, and R&D in the other sample), and we consider their internal accumulated knowledge to be low. Our second group consists of firms that conduct R&I activities occasionally. Their accumulated knowledge is classified as medium. The final category includes firms that persistently engage in renewal efforts resulting in a high level of accumulated internal knowledge. The three categories are labeled  $I_1$ ,  $I_2$  and  $I_3$ , respectively.

For the external knowledge potential of each firm, we identify each firm's access to the supply of knowledge-intensive producer services, which provides a knowledge potential value for every firm<sup>4</sup>. These values allow us to arrange all firms into three categories. The first category includes firms that belong to the lowest third of knowledge potential values. The second is firms that belong to the medium third of knowledge potential values, and the final category consists of firms that belong to the highest third of knowledge potential values. These three categories are labeled  $K_1$ ,  $K_2$  and  $K_3$ .

Based on the two sets of categories, we construct 9 combinatorial categories, as illustrated in Table 1. At one extreme, we find firms with low internal knowledge and low external knowledge potential  $(IK_{11})$ , and the firm at the other extreme has high internal knowledge intensity and high external knowledge potential  $(IK_{33})$ .

|       | $I_1$     | $I_2$     | $I_3$     |
|-------|-----------|-----------|-----------|
| $K_1$ | $IK_{11}$ | $IK_{12}$ | $IK_{13}$ |
| $K_2$ | $IK_{21}$ | $IK_{22}$ | $IK_{23}$ |
| $K_3$ | $IK_{31}$ | $IK_{32}$ | $IK_{33}$ |

Table 1: Combinatorial categories of internal and external knowledge

Before we formulate the hypotheses precisely, we should observe that our formulation enables clarifying the importance of each *IK*-combination. Therefore, we may for example investigate if a strong knowledge potential can compensate for a low level of internal knowledge. We can also determine if firms with persistent R&I efforts can compensate for a low level of external knowledge potential. Thus, we can contribute to the existing literature about the relative importance of the two knowledge factors in Table 1.

<sup>&</sup>lt;sup>4</sup>It should be noted that our knowledge potential indicator also announces the presence of other knowledge sources such as universities, research institutes, high-technology firms and creative capacities.

The first hypothesis refers to the combinatorial categories in the  $I_1$ -row, comprising firms with a low level of internal knowledge. More formally:

**H1**: There is no difference in the TFP growth for firms that belong to  $IK_{11}$ ,  $IK_{21}$  and  $IK_{31}$ , which implies that the local milieu and the external knowledge potential have no additional impact on firms with low internal knowledge.

Our second hypothesis concerns the  $I_2$ -row in Table 1, consisting of firms that make occasional R&D efforts:

**H2**: There is a difference in the TFP growth for firms that belong to the  $I_2$  classification, such that  $IK_{12} < IK_{22} < IK_{32}$ . Thus, the growth rate of firms with occasional R&I is an increasing function of access to external knowledge potential.

The third group of firms is involved in persistent R&I efforts ( $I_3$  firms), and the following hypothesis applies for these firms:

**H3**: There is a difference in the TFP growth for firms that belong to the  $I_3$  classification, such that  $IK_{13} < IK_{23} < IK_{33}$ . Thus, the growth rate of firms with persistent R&D is an increasing function of access to external knowledge potential.

Our remaining hypotheses consider only innovative firms. If such firms have the same external potential, we examine if persistent R&I firms are superior to occasional R&I firms. To accomplish this, we make pairwise comparisons between elements in the  $I_2$  and  $I_3$  columns.

**H4**: Persistent R&D firms have higher TFP growth than firms with occasional R&D efforts, such that  $IK_{13}>IK_{12}$ ,  $IK_{23}>IK_{22}$ ,  $IK_{33}>IK_{32}$ . For all categories of location, there is always a positive improvement on TFP growth from more internal knowledge.

In the next two sections, we present both the data we use to test this analytical model that combines the company's internal and external knowledge and the empirical model that we use to study how different combinations of knowledge affect productivity.

### 3 DATA AND VARIABLES

In our empirical investigation, we use manufacturing and service firm-level data provided by Statistics Sweden. The database contains accounting information on all firms in Sweden, information on the educational background and wages of their employees and the location of the firms.

The analysis applies information about the entire population of firms in the Swedish business sector with at least one employee, and the entire population of employees within these firms in the following ways. First, we calculate the aggregate earnings (wage sum) in each of Sweden's 290 municipalities for all 35 industries that are classified as knowledge-intensive producer services (a list of these industries is provided in the Appendix Table A1). This is our proxy for external knowledge potential. Second, we assign a value of access to potential

knowledge to each firm in the Swedish business sector based on the particular municipality in which they are located. Third, we separate the firm's into three evenly distributed groups based on their potential access to external knowledge.

One-third of each of the approximately 400,000 existing Swedish firms is located in places defined as high accessibility areas and they are concentrated to 25 municipalities. An additional third of these firms are found in 78 municipalities classified as areas with medium access to potential external knowledge, and the remaining firms are located in 187 municipalities with low access to potential external knowledge. With this approach, we capture both the individual firm's proximity to nearby knowledge and the firm's proximity to other firms with similar access to knowledge-intensive producers.

As a second step, we form two panels of firms. In the first panel (i.e., the patent panel), we have matched patent data to the entire population of firms in the Swedish business sector, whereas we match R&D data from the Community Innovation Survey (CIS) in the second panel (i.e., CIS panel). The preferred patent panel is restricted to firms with at least 10 employees on average over the 1997-2008 period. The restriction is motivated by our ambition to compare the empirical analysis using this panel with the same empirical approach applied to CIS data. In Sweden, 10 employees is the lower size limit for participation in the CIS studies.

For the patent panel, we use information from the European Patent Office's worldwide patent statistical database (PATSTAT) complemented with data from the Swedish Patent Office. The panel consists of 40,524 unique firms, approximately 2,000 of which applied for at least one patent between 1997 and 2008. The CIS panel considers only those firms that participated in at least two of three consecutive Community Innovation Surveys (CIS) for 2004, 2006 and 2008. The matched data contain 2,738 unique firms. Both panels are unbalanced, and the second is observed only for the 2000-2008 period.

Using national and international patent applications, we classify firms as persistent innovators, occasional innovators and non-innovators based on observations over the entire 12-year period in the patent panel. An obvious limitation of employing CIS data in a panel setting is that almost all the information pertains only to particular years. One of the few exceptions is the frequency of R&D engagement, where the perspective comprises the most recent three-year period. However, such a period is also too short for the purposes of our research. To extend this information, we construct a data set from three different waves of the CIS survey. In the resulting CIS panel, 40 % of firms are observed in all three surveys, and 60% are observed in two surveys. With overlapping data from the three surveys, we can observe the selected firms' innovation strategies over a 5-7 year period.<sup>5</sup>

 $<sup>^{5}</sup>$ However, the observations for the years 1997-1999 are utilised to obtain lags of the dependent variables. It should be noted that the panel is unbalanced in the sense that we include two voluntary surveys and one compulsory survey, which can cause some selection bias. For instance, the fraction of innovators is 31% in the CIS 2008 data and 54%, on average, in the CIS 2004 and 2006 data.

Table 2 presents summary statistics for the 1997-2008 period, with firms separated into three groups reflecting their long-term innovation strategies. Consider first the patent panel in Columns 1, 3 and 5. If a firm applied for a patent during 6 years or more<sup>6</sup>, we categorise the firm as a persistent innovator. If it applied for a patent in 1-5 years, we consider it an occasional innovator. Firms with no patent applications are non-innovators. The table also reports the corresponding statistics for firms observed in the CIS surveys in Columns 2, 4 and 6. We classify a firm as a persistent innovator for the whole 1997-2008 period if it is reported to be a persistent R&D investor in at least two surveys. Moreover, the firm is classified as non-innovative if it is never reported to be R&D-active. All other firms are considered to be occasional innovators.

In the patent panel, which includes all the approximately 40,000 relevant firms in Sweden, 95% are classified as non-innovative, 4% are classified as occasional innovators and 1% are classified as persistent innovators. In the CIS panel, 45% of firms are defined as non-innovative, 38% are occasional innovators and 17% are persistent innovators.

Consistent with our assumptions based on the literature review in Section 2. the median values of most variables differ for persistently innovative firms compared with firms with no innovation activity or only temporary engagement. Persistently innovative firms are larger than occasionally innovative firms, they have more physical capital, and higher intensities of human capital as well. They are also more likely to belong to multinational groups. Corresponding differences are observed between firms classified as occasionally innovative and non-innovators. With respect to growth rates, the descriptive statistics indicate no differences between the two categories of innovative firms, and the median TFP growth rate is highest for non-innovators. In contrast, only innovative firms grow in size. The table also reveals that persistent innovators are more oriented toward high technology and medium-high technology than other firms. Table 3 displays the distributions of the 66,719 observed patent applications across markets, firm sizes, corporate ownership groups and sectors. The vast majority of patent applications are related to firms with more than 100 employees, a large fraction of which are multinational enterprises (MNEs). Domestic MNEs account for nearly 60 per cent of the applications, and foreign owned MNEs account for 35 per cent. The most patent intensive sectors are high and medium-high technology firms in the manufacturing sector. Knowledgeintensive services are more likely to apply for patents than are low or mediumlow technology manufacturing firms, whereas the opposite is true for other services.

Table 4 reports the transition matrix with year-to-year changes in firms' locations over the two samples. The locations are nearly time-invariant across firms with different innovation strategies and firms with different degrees of access to external knowledge. More than 99 per cent of firms remain in one place over any two consecutive years.

 $<sup>^6{\</sup>rm For}$  a robustness check, 8 years threshold instead of 6 years is also considered. The results are similar.

# 4 EMPIRICAL STRATEGY

To quantify the relationship between TFP and the firm's internal and external knowledge sources, we use an augmented Cobb-Douglas approach specified as a growth model. In doing so, we aim to capture the effect of a particular category of combined knowledge sources on the TFP-growth, conditioned on the growth in the previous period and the TFP-level in the previous period.

Total factor productivity growth is estimated in two steps. Following Levinsohn and Petrin (2003), we first compute TFP as the residual of the Cobb-Douglas production function, where the value added of the firm is the dependent variable and labour inputs (divided into highly educated and ordinary labour), material and physical capital are used as the determinants. In the next step, the growth of TFP is estimated as a function of determinants inside and outside the firm as follows:

$$\Delta lnTFP = \alpha_0 + [I_i \times K_i]\gamma_i + \beta_1 \Delta lnTFP_{i,t-1} + \beta_2 lnTFP_{i,t-1} + (1)$$
  
$$\beta_3 \Delta lnSIZE_{it} + \beta_4 OWNER_{it} + \beta_5 SECTOR_{it} + \mu_i + \tau_t + \varepsilon_{it}$$

where *i* indexes the firm, *t* the year, *I* is a vector of innovation indicators, *K* is a vector of external knowledge indicators,  $\Delta TFP$  is the annual growth rate of total factor productivity, TFP is the level of total factor productivity,  $\Delta SIZE$ is employment growth, and OWNER is corporate ownership. Additionally, the TFP-growth depends on the sector, and we distinguish between six manufacturing and service sectors. The firm and year-specific effects are denoted by  $\mu$ and  $\tau$ , respectively. Finally,  $\varepsilon$  is the idiosyncratic error term.

The key coefficient of interest is  $\gamma_i$ , which determines the response of productivity growth to nine combinatorial categories of internal and external knowledge. It is useful to note that the key variable IK for firm i is almost constant over the period we observe due to the following explanation. First, the *I*-classification is based on the frequency of innovation efforts during the observed period, which means that it does not vary between years. Second, the *K*-classification is based on the knowledge intensity of the firm's location, which is close to 100% identical between year t and year t + 1 according to the transition matrix reported in Table 4.

Based on a procedure suggested by Papke and Wooldridge (2005), we also compute the coefficients and standard errors for long-run effects. The long-run effect is a nonlinear function of the coefficients of the explanatory variables and the lagged dependent variable in Equation (1).

To estimate Equation (1), we use the two-step GMM-system estimator developed by Arellano and Bover (1995) and Blundell and Bond (1998). This approach combines equations in differences of the variables with equations in levels of the variables. The validity of the instruments in the model is evaluated with the Sargan-Hansen test of over-identifying restrictions whereas the Arellano-Bond autoregressive test is used for identifying possible second order serial correlation.

In the GMM-regressions, our preferred model treats the key-variables (sustained innovation strategy)  $\times$  (geographic location) as exogenous due to the characteristics of this variable as discussed above. However, in an alternative regression, we experiment with treating the composite variable for knowledge combination as endogenous. We also report pooled OLS-results for the dynamic model.

### 5 REGRESSION RESULTS

Table 5 presents estimates of Equation (1) using a two-step dynamic GMM estimator with the total factor productivity growth (TFP) as the dependent variable. Two alternative sets of results are displayed in the appendix. Table A2 reports the GMM estimates, and Table A3 reports the OLS estimates. Columns 1 and 2 report short- and long-run estimates for the sample that include the entire population of firms with an average of 10 or more employees over the period 1997-2008, whereas Columns 3 and 4 report the corresponding estimates for the CIS-population, which are restricted to a stratified sample with a firm size of 10 or more employees in the year of the surveys. The first panel is labeled the patent panel, and here we use patent applications as a proxy for innovation activity. The second panel is labeled the CIS panel, and in this case the innovation indicator is R&D-engagement. The key-results are presented in the upper part of the table which is organized in three different panels. In the first panel, rows 1-3 show results for non-innovative firms. In the second rows 4-6 show coefficients for temporary innovators. The third panel presents of TFP-growth with respect to persistently innovative firms in different locations in rows 7-9.

#### 5.1 Basic results

Using  $IK_{11}$  in Table 5 as reference group, the estimates in the first panel are small in absolute value statistically significant only in the first column, showing the GMM-estimates for the typical non-innovative firm located in regions with a medium intensity of external knowledge. In this case innovation is proxied by patent and the sign of the coefficient is negative. The first conclusion, however, is that there are none or almost none growth effects from pure and pecuniary spillovers for non-innovators, regardless of the panel, innovation indicator or estimator we consider.

The table reveals three results about Rows 4-6 and firms temporarily engaged in innovation activities and with low, medium or high accessibility to outside knowledge in the local milieu. First, the estimates are positive and significantly different from the base-group for the patent panel. Second, the growth rate is markedly higher among temporary innovators in milieus where firms have high access to knowledge sources, compared to identical firms in milieus with medium or low access to external knowledge (0.067 versus 0.025 and 0.029, respectively). Third, the CIS-panel results are similar but weaker, with coefficients that are positive but insignificant or only weakly significant. The final important set of results presented in Table 5 concerns the TFP-growth among persistent innovators. Rows 7-9 provide a consistent picture for both samples. First, persistent innovators always have faster TFP growth than other firms, regardless of location. Second, the growth rate for persistent innovators increases with access to external knowledge. Consequently, the size of the estimates is largest for the average persistent R&I firms located in areas with high access to external knowledge. The magnitude of the estimate is 0.221 in the patent sample and 0.159 in the CIS sample.

Table 5 also presents the long-run estimates for the two samples, given in Columns 2 and 4, and these results are by default fully consistent with the short-run estimates in Columns 1 and 3. Examining the covariates displayed in Table 5, we find negative signs for both TFP-growth and TFP-level in the previous year. While the latter indicates a tendency to convergence in line with predictions from growth theory, the former deserve some comments. Why is growth in a given year, a negative function of last year's growth rate in our data? There might be a possibility that firms in general simply follow a quiet-life behavior pattern. Hence, the improvement in the performance yesterday reduces the incentives for firms to invest their efforts in better performance(growth) today. Instead they decide to enjoy fruits of their earlier activities. For a discussion on similar findings, see Hashi and Stojcic (2013).

In contrast to the preferred patent-panel, the alternative CIS-panel contains a smaller proportion of firms with falling productivity, because the panel is a selected group of surviving firms over a 4-7 year period. This difference between the panels is also reflected in the TFP estimate for the CIS-panel, which is indeed negative but close to zero (0.036) and insignificant. Turning to other controls, one noteworthy but not unexpected result is that multinational firms have a higher growth rate than other firms, *ceteris paribus*. The TFP growth is notably neutral with respect to firm size, even after controlling for internal and external knowledge.

The test statistics are reported in the lower part of Table 5. We use lag limits t-3 instruments for the regression in differences in both panels and lagged differences dated t-1 for the regression in levels in the patent panel and t-2 in the CIS panel. This results in 93 instruments in the patent panel regression and 81 instruments in the CIS panel regressions, which are both within a reasonable range. The AR(2) test rejects the presence of second-order autocorrelation in the first-differenced residuals in both regressions. Otherwise the GMM estimator could be inconsistent. The Hansen J-test of over-identifying restrictions confirms that the instruments are valid, and the difference-in-Hansen test confirms that the additional instruments required for systems estimation are valid for the two regressions.

#### 5.2 Wald test of the predictions

Overall, the results in Table 5 indicate a strong, positive relationship between proximity to knowledge and persistent R&I (innovation activities measured by patent or R&D). To evaluate the quantitative importance of the IK coefficients in detail, we conduct a Wald test on the equality of means in Table 6. The first prediction from our Hypotheses H1-H4 is that the local milieu and the external knowledge potential have no additional impact on firms with low internal knowledge. The H1 section of the table indicates that non-innovators in places with medium access to knowledge outside the firm have only somewhat lower growth rates than the reference group (non-innovators in locations with low access to external knowledge) in the patent panel. No significant difference is found in the CIS panel. We therefore confirm Hypothesis 1.

Our second prediction, that the growth rate of firms with occasional R&I is an increasing function of access to external knowledge, is partly confirmed when the patent panel is considered in the H2 portion of Table 6. Temporary innovators in high-access areas are growing faster than temporary innovators located in other places. However, no significant difference exists in the equality of growth means between firms that are occasionally engaged in places with low and medium access to external knowledge. We also find no significant difference in the coefficients with respect to location in the CIS panel. Thus, we cannot confirm hypothesis H2 based on the CIS-panel and only partly when we use the patent-panel.

We turn to the prediction that the growth rate of firms with persistent R&I is an increasing function of access to external knowledge (H3). The results for the patent panel indicate that persistent innovators in high access (knowledge) regions are growing significantly faster than corresponding firms in both mediumand low-access regions. Moreover, persistent innovators in medium-access places have higher growth rates than persistent innovators in low-access locations.

The CIS panel indicates that persistent innovators in areas with high access to external knowledge are growing significantly faster than the corresponding firms with low access to external knowledge. However, we cannot conclude that the estimation for persistent innovators in places with high access to external knowledge (0.159) is significantly larger than the coefficient for persistent innovators in locations with medium access (0.123) or that the medium-access estimate is greater than the estimate for low access (0.094). The overall assessment based on the estimation of the two panels is that we cannot reject the third hypothesis.

Our final prediction (H4) is that a positive return to improvement of internal knowledge always applies for all categories of location, which implies that persistent innovators outperform occasional innovators in all types of locations. The prediction is strongly confirmed in both of our panels.

#### 5.3 Sensitivity test

One concern with the results presented in Table 4 is the possibility that the key-variable IK might be considered to be endogenous and that an exogenous shock might affect both the TFP-growth of the firms and their choice of location. Although the transition matrix reported in Table 4 indicates that the likelihood of a firm moving between our three location categories is very low, this concern is partially addressed in Table A2 by rerunning Equation (1) and dropping the exogenous assumption on the interaction-variable IK. However, the instrument for the IK-variable is not meaningful because by construction the variables are almost completely identical to the variable itself. Despite this methodological limitation, in Table A2 we undertake an analysis of the TFP-growth using the endogeneity assumption of the nine IK-variables.

The key results in Table A2 are changed substantially as a result of the different treatment of the IK-variable. The estimates for non-innovators in places with medium and high access to external knowledge and for occasional innovators across all regions as well as the coefficients for persistent innovators in regions with low access to external knowledge are not significantly different from the reference group. In sharp contrast, the estimates for persistent innovators in locations with medium and high access to knowledge are considerably larger than for those obtained from the preferred GMM-specification. These two estimates are large and significant in both the patent panel and the CIS panel.

The most marked difference in the GMM results, depending on whether we are treating the IK variable as exogenous or endogenous, obtains for the occasional innovators. While Table 5 shows some limited influence from the surrounding local environment, Table A2 reports that these estimates are insignificant. This result also applies to persistent innovative companies in geographic environments with low external knowledge intensity.

Table A3 in the appendix indicates that the pooled OLS estimates of Equation (1) produce the same overall results as the more efficient dynamic GMM estimates<sup>7</sup>. The main difference is the sizes of the estimates, which are lower when using the OLS estimator. The estimator suffers from dynamic panel data bias as well as bias due to serial correlation in the error term and potential endogeneity.

What then are the common observations in the three tables? Table 5 and Tables A2 and A3 in the appendix reveal four regularities that persist in alternative specifications and estimators. First, the differences in the coefficient estimates among non-innovators in different locations are negligible. Second, our evidence that occasionally innovative companies grow faster in a knowledge-intensive environment is weak. Third, the growth rates for persistently innovative firms in locations with high access to external knowledge are always higher than those of firms in other locations, regardless of innovation strategy. Finally, for persistent

<sup>&</sup>lt;sup>7</sup>A fixed-effect model is used to estimate the lag of the dependent variable for all regressions. The results indicate that the coefficients on lagged dependent variables using the GMM estimator are higher than the coefficients obtained for the fixed effect model and lower than the OLS estimates. The results are available upon request.

innovators, the growth rates are always increasing with the amount of external knowledge.

Although all three regressions suggest economically important effects of internal and external knowledge on TFP-growth for persistent innovators, only the preferred GMM-model and the OLS estimates also indicate positive effects for temporary innovators in places with high access to external knowledge. However, the latter finding is only relevant for the patent sample. Overall, proximity appears to be more important for innovative firms, consistent with our a priori assumption.

# 6 CONCLUSIONS

Our study aims to illuminate the separate and combined effect of innovation and potential spillovers from a growth perspective. A significant amount of prior research supports the view that (i) a firm's knowledge is a key competitive asset (Grant, 1996), (ii) continuity of innovation efforts ensures the accumulation of internal knowledge (Dosi and Nelson, 2010), (iii), very few firms, if any, can internally develop all critical knowledge needed for growth (Almeida and Phene, 2012), (iv) a firm's potential for exploiting external knowledge and recombining internal and external knowledge increases with its own knowledge stock (Cohen and Levinthal, 1989), and (v) locational proximity to external knowledge reduces the cost and increases the frequency of contacts with players in a network (Saxenian,1990). Building on these and similar findings, we construct a simple analytical model that examines how firms exploit internal knowledge in conjunction with external knowledge to gain productivity growth.

We model knowledge inputs in a production function by using a discrete composite variable with nine different combinations of the intensity of knowledge from within and from outside the firm. Internal knowledge is measured by the frequency of national and international patent applications. We have matched patent applications to all 40,524 unique manufacturing and service firm in Sweden with an average of 10 or more employees from 1997 to 2008. A second alternative panel is constructed from an overlapping data set from three Swedish Community Innovation Surveys in 2004, 2006 and 2008 for which 2,738 manufacturing and service firms that participated in at least two of the three surveys. In this case, the data are restricted to firms with at least 10 employees during the year of the survey.

To find a proxy for knowledge flow across firms, we identify 35 different Swedish knowledge intense producer-service industries at the 5-digit level in which the share of employees with university degree is above 30 percent. These services include ICT services, engineering R&D and engineering services, financial services, and brokerage and recruitment of manpower.

Applying a dynamic GMM-estimator to the data, which also includes extensive firm characteristics on human capital, physical capital, employment, ownership and sector classification, two equations were estimated. The main findings are as follows:

- The local milieu and the external knowledge potential have no additional productivity growth impact on firms with low internal knowledge.
- The growth rate of total productivity is only weakly associated with external knowledge for firms with occasional innovation efforts.
- The growth rate of total productivity is strongly associated with external knowledge for firms with persistent innovation efforts.
- All location categories exhibit improvement of internal knowledge.

Our study provides new empirical knowledge about the systematic differences of firms' capability to benefit from external knowledge. It also suggests a method for capturing and quantifying the extent of knowledge flows across firms. Moreover, the study demonstrates the appropriateness of using the increasingly popular dynamic GMM-estimator to control whether productivity and growth results are due to observed heterogeneous characteristics of firms and places or factors such as unobserved heterogeneities or true or false state dependence. Our work also complements to some recent studies.

The above findings have implications for both policy and management. With our approach the results indicate that the benefits of knowledge-intensive local milieus are not uniformly distributed across different types of firms. We find strong effects on TFP growth only for innovating firms and especially for persistent innovators. We do not detect any substantial effect for occasional innovators and no effect at all for non-innovators, which constitute the vast majority of all firms. Thus, while the policy debate tends to assume that firms located in knowledge-rich milieus such as urban agglomerations and specialized spatial clusters will profit from proximity to diversified knowledge and supply of knowledge-intensive producer services, in technology, law, finance, management, marketing and other support functions, the study contributes to a more nuanced discussion. Our distinct results support recent studies suggesting that policymakers and managers should not expect that the presence of a knowledge-intensive environment automatically leads to leverage effects on firm performance. Instead, supportive innovation policies should consider measures that help to maintain and improve the knowledge milieu of places in which many firms follow strategies that give priority to a permanent innovation engagement. The result from our study also raises the complex question: which policies can facilitate the transition of a firm from a state of being an occasional innovator to being persistently engaged in innovation efforts. Occasional efforts include disruptions that can cause the erosion and obsolescence of acquired skills, routines and technology. The policy nexus of our study is two-pronged. A firm's knowledge management comprises (i) systematic accumulation of internal knowledge combined with the development of absorption and accession capacity, and (ii) location in knowledge-intensive environment. The basic policy message is that these two components are not substitutes, but rather complements.

There are several limitations of this study that can be questions for future research. First, the issue of knowledge flows across firms that are not related to links within the nearby milieu of the firms is not explicitly addressed in this paper, except for the effect associated with multinational company groups. Recently Cantwell and Piscitello (2015) have used openness of the regional industry and the regional economy to capture global knowledge diffusion, while other papers apply methods such as trade statistics, patent citations and strategic alliances. A second issue that deserves a more subtle analysis than what is provided in the present paper is the internal mechanisms for creating and maintaining conduits to the external environment that facilitates knowledge flows to the firm. Another issue for future research is to investigate the importance of the corporate ownership. Are multinational firms more efficient to exploit external local knowledge than other firms? Is there any difference in the ability to benefit from the nearby milieu between domestically owned firms and foreign firms?

## 7 References

# References

- Almeida, P., (1996), Knowledge sourcing by foreign multinationals: Patent citation analysis in the U.S. semiconductor industry, *Strategic Management Journal*, Special Issue 7, 155-165.
- [2] Almeida, P., Phene, A., (2012), Managing Knowledge Within and Outside the Multinational Corporation, in Innovation and Growth: From R&D Strategies of Innovating Firms to Economy-wide Technological Change (1rst Ed. Pp. 21-37), Oxford University press.
- [3] Arora A, Gambardella A., (1994), Evaluating Technological Information and Utilizing It – Scientific Knowledge, Technological Capability, and External Linkages in Biotechnology. Journal of Economic Behavior & Organization 24(1), 91-114.
- [4] Arrow, K., (1962). Economic Welfare and the Allocation of Resources for Invention. In: R. Nelson (ed.) The Rate and Direction of Inventive Activity: Economic and Social Factors, 609-626. New York: National Bureau of Economic Research, Inc.
- [5] Attewell, P., (1992). Technology Diffusion and Organizational Learning: The Case of Business Computing, Organizational Science 3, 1–19.
- [6] Arellano, M., Bover, O., (1995). Another Look at Instrumental Variable Estimation of Error Component Models. *Journal of Econometrics*, 68, 29-51.
- [7] Bartelsman, E., Doms, M., (2000). Understanding productivity: Lessons from longitudinal microdata. *Journal of Economic Literature* 38, 569–594.
- [8] Beckmann, M.J., (2000). Interurban Knowledge Networks. In: Batten D (ed), Learning, Innovation and Urban Evolution, Kluwer Academic, London, 127-135.
- [9] Belderbos, R., Duysters, G., & Sabidussi, A. (2012). R&D collaboration and Innovative performance., in Innovation and Growth: From R&D Strategies of Innovating Firms to Economy-wide Technological Change (1rst Ed. Pp. 160-181), Oxford University press.
- [10] Blundell, R.S., Bond, S., (1998). Initial Conditions and Moment Restrictions in Dynamic Panel data Models. *Journal of Econometrics* 87(1), 115-43.
- [11] Breschi, S., Lissoni, F., (2001). Knowledge spillovers and local innovation systems: a critical survey. *Industrial and Corporate Change* 10, 975-1005.

- [12] Brown, J. R., Fazzari, S. M., Petersen, B. C., (2009). Financing innovation and growth: cash-flow, external equity, and the 1990s R&D boom. *Journal* of Finance, 64, 151-185.
- [13] Brouwer, E., Nadvornikova H.B., Kleinknecht, A., (1999). Are Urban Agglomerations a Better Breeding Place for Product Innovations? An Analysis of New Product Announcements, *Regional Studies* 33, 541–549
- [14] Cantwell, J., Piscitello, L. (2015). New competence creation in MNC subunits: The role of international knowledge, *The World Economy*, forthcoming
- [15] Cassiman, B., Veugelers, R., (2002). R&D cooperation and spillovers: some empirical evidence from Belgium. *American Economic Review* 92 (4), 1169–1184.
- [16] Cassiman, B., Veugelers, R., (2006). In search of complementarity in innovation strategy: internal R&D and external knowledge acquisition. *Man-agement Science* 52 (1), 68–82.
- [17] Cefis, E., Orsenigo, L. (2001). The persistence of innovative activities: A cross-countries and cross-sectors comparative analysis. *Research Policy* 30 (7), 1139-1158
- [18] Ciccone, A., Hall, R.E., (1996). Productivity and the Density of Economic Activity, NBER Working Papers 4313.
- [19] Cohen, W., Levinthal, D., (1990) Absorptive capacity: A new perspective on learning and innovation. Administrative Science Quarterly 35(1), 128-158.
- [20] Cubbin, J., Geroski, P. (1987). The convergence in profits in the long-run: Inter-firm and inter-industry comparisons. *Journal of Industrial Economics* 35, 427–442.
- [21] Doloreux, D., Shearmur, R. (2012). Collaboration, information and the geography of innovation in knowledge intensive business services. *Journal* of economic geography, 12(1), 79-105.
- [22] Dosi, G. (2007). Statistical regularities in the evolution of industries. A guide through some evidence and challenges for the theory. In: Malerba, F., Brusoni, S. (Eds.), Perspectives on Innovation. Cambridge University Press, Cambridge.
- [23] Dosi, G., Nelson, R.R., (2010). Technical Change and Industrial Dynamics as Evolutionary Processes. In B. Hall and N. Rosenberg (eds.) The Economics of Innovation, Elsevier.
- [24] Duguet, E., Monjon, S., (2002). Les fondements microéconomiques de la persistance de l'innovation. Une analyse économétrique, Revue Économique, Programme National Persée 53(3), 625-636.

- [25] Escribano, A., Fosfuri, A. Tribo, J.A., (2009). Managing external knowledge flows: the moderating role of absorptive capacity, *Research Policy* 38 (1), 96–105.
- [26] Frenken, K., van Oort, F. G., and Verburg, T. (2007). Related variety, unrelated variety and regional economic growth. *Regional Studies* 41:685–97.
- [27] Fujita, M., Thisse, J.F (2002). Economics of agglomeration: Cities, industrial location and regional growth. Cambridge: Cambridge University Press
- [28] Geroski, P.A., (1998). An applied econometrician's view of large company performance. *Review of Industrial Organization* 13, 271–293.
- [29] Geroski, P.A., Jacquemin, A., (1988). The persistence of profits: A European comparison. *Economic Journal* 98, 357–389.
- [30] Glaeser, E. L., & Gottlieb, J. D. (2009). The wealth of cities: Agglomeration economies and spatial equilibrium in the United States, National Bureau of Economic Research., Working Paper 14806
- [31] Grant, R. M. (1996). Toward a knowledge-based theory of the firm. Strategic management journal, 17, 109-122.
- [32] Gschwandtner, A., (2004). Profit persistence in the "very" long run: Evidence from survivors and exiters. Vienna Economics Papers 0401, University of Vienna, Department of Economics.
- [33] Hagedoorn, J. (2002). Inter-firm R&D partnerships: an overview of major trends and patterns since 1960. *Research policy*, 31(4), 477-492.
- [34] Hashi, I., Stojčić, N. (2013). The impact of innovation activities on firm performance using a multi stage model: Evidence from the Community Innovation Survey 4. *Research Policy*, 42(2), 353-366.
- [35] Huber, F., (2012). Do clusters really matter for innovation practices in Information Technology? Questioning the significance of technological knowledge spillovers. *Journal of Economic Geography*, 12, 107-126.
- [36] Jacobs, J., (1969). The Economy of Cities. New York; Vintage. Kogut, B., Zander, U., 1992. Knowledge of the firm, combinative capabilities, and the replication of technology. Organization Science 3 (3), 383-397.
- [37] Klette, T.J., Kortum, S., (2004). Innovating Firms and Aggregate Innovation, Journal of Political Economy 112, 986-1018.
- [38] Levinsohn, J., Petrin, A., (2003). Estimating Production Functions Using Inputs to Control for Unobservables, *The Review of Economic Studies*, 70(2), 317-341

- [39] Lychagin, S., Pinkse, J., Slade, M.E., Van Reenen. J., (2010). Spillovers in Space: Does Geography Matter?, NBER Working Papers 16188, National Bureau of Economic Research.
- [40] Lööf, H., Johansson, B. (2014) R&D Strategy, Metropolitan Externalities and Productivity: Evidence from Sweden. *Industry and Innovation* 21(2).
- [41] Marshall, A., (1890). Principles of Economics. London: Macmillan and Co.
- [42] Maurseth, P. B., Verspagen, B., (1999). Knowledge Spillovers in Europe. A Patent Citation Analysis,' paper presented at the CRENOS Conference on Technological Externalities and Spatial Location, University of Cagliari, 24–25 September.
- [43] Mueller, D. (1986). Profits in the Long-Run. Cambridge University Press, Cambridge.
- [44] Nelson R., Winter, S.G., (1982). An Evolutionary Theory of Economic Change. Cambridge, Mass: The Belknap of Harvard University Press.
- [45] Nobel, R., Birkinshaw, J., (1998). Innovation in multinational corporations: Control and communication patterns in international R&D operations, *Strategic Management Journal* 19(5), 479-496.
- [46] Nooteboom, B., Van Haverbeke, W., Duysters, G., Gilsing, V., & Van den Oord, A. (2007). Optimal cognitive distance and absorptive capacity. *Research Policy*, 36(7), 1016-1034.
- [47] Papke, L., Wooldridge, J., (2005). A Computational Trick for Delta-Method Standard Errors. *Economics Letters* 86 (3), 413-417.
- [48] Peters, B., (2009). Persistence of innovation: stylized facts and panel data evidence. Journal of Technology Transfer 34, 226-243.
- [49] Peters, B., Roberts, M. J., Vuong, V. A., & Fryges, H. (2013). Estimating dynamic R&D demand: An analysis of costs and long-run benefits (No. w19374). National Bureau of Economic Research.
- [50] Phene, A., Almeida, P., (2008). Innovation in multinational subsidiaries: The role of knowledge assimilation and subsidiary capabilities, *Journal of International Business Studies* 39, 901–919.
- [51] Polanyi, M., (1966). The Tacit Dimension. Doubleday & Company, New York.
- [52] Porter, M., (1990). The Competitive Advantage of Nations. New York: Free Press.
- [53] Raymond, W., Mohnen, P., Palm, F., & van der Loeff, S. S. (2010). Persistence of innovation in Dutch manufacturing: Is it spurious?. *The Review* of Economics and Statistics, 92(3), 495-504.

- [54] Romer, P.M., 1986. Increasing Returns and Long-Run Growth. Journal of Political Economy 94, 1002-1037
- [55] Rosenberg, N., (1990). Why do firms do basic research (with their own money)? Research Policy 19, 165–174.
- [56] Rosenthal, S. S., & Strange, W. C. (2003). Geography, industrial organization, and agglomeration. review of Economics and Statistics, 85(2), 377-393.
- [57] Saxenian, A., (1990). Regional networks and the resurgence of Silicon Valley, *California Management Review*, Fall, 39-112.
- [58] Syverson, C., (2004). Product substitutability and productivity dispersion. *Review of Economics and Statistics* 86 (2), 534–550.
- [59] Syverson, C., (2011). What Determines Productivity? Journal of Economic Literature 49(2), 326–365
- [60] Teece, D.J., (2010). Technological Innovation and the Theory of the Firm: The Role of Enterprise-Level Knowledge, Complementarities, and (Dynamic) Capabilities. In Eds. B. Hall and N. Rosenberg Handbook of the Economics of Innovation, Elsevier North-Holland, 2010
- [61] Wolff, E., (2012). Spillover, Linkages, and Productivity Growth in the US Economy, 1958 to 2007. In Andersson, M., Karlsson, C., Johansson, B., Lööf, H (Eds) Innovation and Growth: From R&D Strategies of Innovating Firms to Economy-wide Technological Change. Oxford University Press.
- [62] Verspagen, B., Schoenmakers, W., 2000. The Spatial Dimension of Knowledge Spillovers in Europe: Evidence from Patenting Data, paper presented at the AEA Conference on Intellectual Property Econometrics, Alicante, 19–20 April.
- [63] von Hippel, E. (1994). Sticky Information and the Locus of Problem Solving: Implications for Innovation. *Management Science* 40(4): 429-439.
- [64] Åstebro, T., (2002). Noncapital Investment Costs and the Adoption of CAD and CNC in U.S. Metal working Industries, *Rand Journal of Economics* 33(4), 672–8.

# 8 Tables

| Table 2: Descriptive statistics for 1997-2008. | Innovation strategy | based on patent | applications |
|--|---------------------|-----------------|--------------|
| and the CIS-panel (median and standard erro    | ors)                |                 |              |

|  |                         | (1)        |        | (2)       |                   | (3)       |
|--|-------------------------|------------|--------|-----------|-------------------|-----------|
|  | Nor                     | n R&I      | Occasi | ional R&I | $\mathbf{Persis}$ | tent R&I  |
|  | $\operatorname{Patent}$ | CIS panel  | Patent | CIS panel | Patent            | CIS panel |
| TFP growth $^{a,c}$                              | 0.032                   | 0.028      | 0.025  | 0.026     | 0.025             | 0.023     |
|  | (0.47)                  | (0.39)     | (0.49) | (0.39)    | (0.53)            | (0.44)    |
| Human capital $^{b}$                             | 0.036                   | 0.026      | 0.071  | 0.053     | 0.145             | 0.111     |
|  | (0.17)                  | (0.14)     | (0.19) | (0.17)    | (0.21)            | (0.21)    |
| Firm size $^{a}$                                 | 2.83                    | 2.94       | 3.56   | 3.43      | 4.73              | 4.75      |
|  | (0.98)                  | (1.22)     | (1.30) | (1.38)    | (1.62)            | (1.75)    |
| Firm size growth                                 | 0                       | 0          | 0.013  | 0.005     | 0.012             | 0.012     |
|  | (0.38)                  | (0.29)     | (0.31) | (0.29)    | (0.28)            | (0.26)    |
| Physical capital $^{a,c}$                        | 13.69                   | 14.20      | 14.82  | 14.80     | 16.28             | 16.32     |
|  | (2.89)                  | (2.96)     | (2.62) | (2.65)    | (2.75)            | (2.75)    |
| Domestic Non Affiliate Firms                     | 0.44                    | 0.40       | 0.20   | 0.29      | 0.09              | 0.14      |
| Domestic Uninational Firms                       | 0.34                    | 0.32       | 0.23   | 0.29      | 0.09              | 0.17      |
| Domestic Multinational Firms                     | 0.12                    | 0.14       | 0.36   | 0.21      | 0.47              | 0.40      |
| Foreign Multinational Firms                      | 0.10                    | 0.14       | 0.21   | 0.21      | 0.35              | 0.29      |
| High tech manufact <sup><math>b</math></sup>     | 0.01                    | 0.04       | 0.07   | 0.06      | 0.17              | 0.18      |
| Medium-High tech manu <sup>b</sup>               | 0.05                    | 0.11       | 0.27   | 0.18      | 0.36              | 0.28      |
| Medium-Low tech manu <sup>b</sup>                | 0.08                    | 0.15       | 0.21   | 0.18      | 0.16              | 0.15      |
| Low tech manu <sup><math>b</math></sup>          | 0.10                    | 0.24       | 0.12   | 0.27      | 0.07              | 0.17      |
| Knowledge-intense serv <sup><math>b</math></sup> | 0.27                    | 0.17       | 0.14   | 0.16      | 0.13              | 0.17      |
| Other $\operatorname{serv}^b$                    | 0.46                    | 0.26       | 0.18   | 0.13      | 0.10              | 0.05      |
| Mining $^{b}$                                    | 0.03                    | 0.03       | 0.01   | 0.00      | 0.01              | 0.00      |
| Observations total                               | $296,\!105$             | $12,\!926$ | 13,323 | 10,841    | 4,067             | 4,837     |
| Unique   | 38,703                  | 1,246      | 1,422  | 1,025     | 399               | 467       |
| Observations, fraction                           | 0.95                    | 0.45       | 0.04   | 0.38      | 0.01              | 0.17      |

Note: a)Log, b)Fraction, c)Real prices

|                                | Number of Applications | OccasionalR&I, % | PersistentR&I, % |
|--------------------------------|------------------------|------------------|------------------|
| Knowledge access: Low          | 6,947                  | 0,25             | 0,75             |
| Knowledge access: Medium       | $31,\!089$             | 0,05             | $0,\!95$         |
| Knowledge access: High         | $28,\!590$             | 0,06             | 0,94             |
| 10-25                          | 3,308                  | $0,\!47$         | 0,53             |
| 26-99                          | 5,860                  | 0,32             | $0,\!68$         |
| 100>                           | $57,\!458$             | 0,03             | $0,\!97$         |
| Domestic Non Affiliate Firms   | 2,427                  | 0,39             | 0,61             |
| Domestic Uninational Firms     | 2,301                  | $0,\!37$         | $0,\!63$         |
| Domestic Multinational Firms   | 38,364                 | 0,05             | 0,95             |
| Foreign Multinational Firms    | $23,\!534$             | 0,05             | 0,95             |
| High tech manufacturing        | 31,572                 | 0,02             | 0,98             |
| Medium-High tech manufacturing | 16,361                 | $0,\!10$         | $0,\!90$         |
| Medium-Low tech manufacturing  | $5,\!510$              | $0,\!15$         | 0,85             |
| Low tech manufacturing         | 3,549                  | $0,\!14$         | 0,86             |
| Knowledge-intense services     | 7,202                  | $0,\!12$         | $0,\!88$         |
| Other services                 | 2,339                  | 0,35             | $0,\!65$         |
| Mining                         | 93                     | $0,\!22$         | 0,78             |

Table 3: Distribution of the patent applications during the 1997-2008 period by firms in Sweden across regions and groups

 Table 4: Transition Matrix

|              | Access to external knowledge | No R&I,% | Occasional R&I,% | Persistent R&I,% |
|--------------|------------------------------|----------|------------------|------------------|
| Patent panel | Low                          | 99.3     | 99.6             | 99.1             |
|              | Medium                       | 99.1     | 99.1             | 99.3             |
|              | $\operatorname{High}$        | 99.4     | 98.9             | 99.0             |
| CIS panel    | Low                          | 99.5     | 99.5             | 99.6             |
|              | Medium                       | 99.5     | 99.1             | 99.5             |
|              | $\operatorname{High}$        | 99.4     | 98.9             | 99.5             |

Note: The matrix shows that all firms in all three categories of geographical areas tend to remain in the same place across time.

| Innovation variable                                  | PATENT        | PANEL         | CIS P.        | ANEL         |
|--|---------------|---------------|---------------|--------------|
|  | Short-run     | Long-run      | Short-run     | Long-run     |
| IK <sub>11</sub> <sup>a</sup>                        | 0.000         | 0.000         | 0.000         | 0.000        |
| $IK_{12}$  | -0.008**      | -0.01**       | -0.015        | -0.02        |
|  | (0.004)       | (0.00)        | (0.011)       | (0.01)       |
| IK <sub>13</sub>                                     | 0.002         | 0.00          | 0.003         | 0.00         |
|  | (0.003)       | (0.00)        | (0.012)       | (0.01)       |
| IK <sub>21</sub>                                     | 0.029 * * *   | $0.02^{***}$  | 0.016         | 0.02         |
|  | (0.009)       | (0.01)        | (0.011)       | (0.01)       |
| $IK_{22}$  | $0.025^{**}$  | 0.02**        | 0.017         | 0.02         |
|  | (0.010)       | (0.01)        | (0.011)       | (0.01)       |
| IK <sub>23</sub>                                     | $0.067^{***}$ | $0.06^{***}$  | 0.031*        | $0.03^{*}$   |
|  | (0.017)       | (0.01)        | (0.017)       | (0.02)       |
| IK <sub>31</sub>                                     | 0.082***      | 0.07***       | 0.094***      | 0.09***      |
|  | (0.026)       | (0.02)        | (0.035)       | (0.03)       |
| IK <sub>32</sub>                                     | $0.126^{***}$ | $0.11^{***}$  | $0.123^{***}$ | $0.12^{***}$ |
|  | (0.034)       | (0.03)        | (0.040)       | (0.04)       |
| IK <sub>33</sub>                                     | 0.221***      | $0.19^{***}$  | 0.159 * * *   | $0.15^{***}$ |
|  | (0.054)       | (0.05)        | (0.043)       | (0.04)       |
| Log Firm size, growth                                | 0.021         | 0.05          | -0.012        | 0.01         |
|  | (0.072)       | (0.07)        | (0.122)       | (0.13)       |
| Log TFP growth $t-1$                                 | -0.147**      |               | -0.036        |              |
|  | (0.073)       |               | (0.071)       |              |
| $\log \mathrm{TFP}_{t-1}$                            | -0.239***     |               | -0.383***     |              |
|  | (0.066)       |               | (0.107)       |              |
| Domestic Uninational <sup><math>b</math></sup>       | $0.033^{**}$  | $0.03^{**}$   | 0.023         | 0.022        |
|  | (0.013)       | (0.011)       | (0.017)       | (0.015)      |
| Domestic multinational <sup><math>b</math></sup>     | $0.083^{***}$ | $0.07^{***}$  | $0.126^{***}$ | $0.12^{***}$ |
|  | (0.028)       | (0.02)        | (0.046)       | (0.042)      |
| For eign multination $a^b$                           | 0.099 * * *   | $0.086^{***}$ | $0.145^{***}$ | $0.14^{***}$ |
|  | (0.032)       | (0.03)        | (0.051)       | (0.05)       |
| Observations   | 196,027       |               | 20,076        |              |
| Unique firms   | $31,\!208$    |               | $2,\!634$     |              |
| Laglimits  | (3 1)         |               | $(3\ 2)$      |              |
| Instruments  | 93            |               | 81            |              |
| AR(2)  | 0.914         |               | 0.206         |              |
| Hansen Overid.                                       | 0.231         |               | 0.178         |              |
| Difference-in-Hansen test for level equation         | 0.846         |               | 0.192         |              |
| Difference-in-Hansen test for lag dependent variable | 0.299         |               | 0.107         |              |

Table 5: Dependent variable: TFP growth, Two-step system GMM Estimates

Note: \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%

(a) Reference group (b) Reference group is domestic non-affiliated firms

Robust (GMM) standard error in parentheses. Year and sector dummies included

[IK11: Non R&I and Low access]; [IK12: Non R&I and Medium access]; [IK13: Non R&I and High access];

[IK21: Occ. R&I and Low access]; [IK22: Occ. R&I and Medium access]; [IK23: Occ. R&I and High access]

[IK<sub>31</sub>: Pers. R&I and Low access]; [IK<sub>32</sub>: Pers. R&I and Medium access]; [IK<sub>33</sub>: Pers. R&I and High access] 25

|                                    | Hypotheses | Patent panel  | CIS panel     |
|------------------------------------|------------|---------------|---------------|
|                                    |            | t-test        | t-test        |
| IK <sub>13</sub> =IK <sub>12</sub> | H1         | 0.003***      | 0.153         |
| $IK_{13} = IK_{11}$                | H1         | 0.008**       | 0.015         |
| $IK_{23} = IK_{22}$                | H2         | 0.004***      | 0.366         |
| $IK_{23} = IK_{21}$                | H2         | $0.006^{***}$ | 0.323         |
| $IK_{22} = IK_{21}$                | H2         | 0.667         | 0.983         |
| IK <sub>33</sub> =IK <sub>32</sub> | H3         | 0.008***      | 0.177         |
| $IK_{33} = IK_{31}$                | H3         | 0.000 * * *   | 0.020**       |
| $IK_{32} = IK_{31}$                | H3         | 0.039**       | 0.269         |
| $IK_{33} = IK_{23}$                | H4         | 0.000***      | 0.000***      |
| $IK_{32} = IK_{22}$                | H4         | 0.000 * * *   | $0.006^{***}$ |
| $IK_{31} = IK_{21}$                | H4         | 0.017**       | 0.015**       |

Table 6: T-test on the equality of means reported as p-values

Note: The table report t-test for hypotheses H1-H4.

P-values and degrees of significance are reported.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%

[IK<sub>11</sub>: Non R&I and Low access]; [IK<sub>12</sub>: Non R&I and Medium access]; [IK<sub>13</sub>: Non R&I and High access]
[IK<sub>21</sub>: Occ. R&I and Low access]; [IK<sub>22</sub>: Occ. R&I and Medium access]; [IK<sub>23</sub>: Occ. R&I and High access]
[IK<sub>31</sub>: Pers. R&I and Low access]; [IK<sub>32</sub>: Pers. R&I and Medium access]; [IK<sub>33</sub>: Pers. R&I and High access]

# A Appendix

| SIC 2002 | Industry  | Knowledge    | Fraction   |
|----------|---|--------------|------------|
|          |   | intensity, % | of KIPS 30 |
| 7220     | Software consultancy and supply                                   | 46,1         | 18,45      |
| 74202    | Construction and other engineering activities                     | 38,4         | 16,84      |
| 65120    | Monetary intermediation   | 32,5         | $12,\!28$  |
| 74140    | Business and management activities                                | 45,2         | 11,16      |
| 74120    | Accounting, book-keeping and auditing activities: tax consultancy | 41,2         | 7,71       |
| 72210    | Publishing of software  | $50,\!3$     | 5,13       |
| 74501    | Labor recruitment activities                                      | 35,9         | 3,98       |
| 73102    | R&D on engineering and technology                                 | 68,5         | 3,15       |
| 74111    | Legal advisory  | 70,9         | 2,45       |
| 74850    | Secretarial and translation activities                            | 32,9         | 2,00       |
| 65220    | Credit granting   | 31,7         | 1,90       |
| 61102    | Sea and costal water transport                                    | 42,8         | 1,90       |
| 74201    | Architectural activities  | 67,1         | 1,84       |
| 73103    | R&D medical and pharmaceutical science                            | 69,7         | 1,50       |
| 73101    | R&D on natural science  | 74,3         | 0,97       |
| 74104    | R&D on agricultural science                                       | 67,1         | 0,92       |
| 74130    | Market research and public opinion pulling                        | 36,1         | 0,87       |
| 74872    | Design activities   | 32,4         | 0,86       |
| 67120    | Security broking and fund management                              | 52,7         | 0,84       |
| 66012    | Life insurance  | 33,8         | 0,79       |
| 67202    | Activities auxiliary to insurance and pension funding             | 31,6         | 0,74       |
| 72400    | Data base activities  | 31,7         | 0,70       |
| 65232    | Unit trust activities   | 36,5         | 0,58       |
| 65231    | Investment trust activities                                       | 49,7         | 0,53       |
| 74112    | Advisory activities concerning patents and copyrights             | 50,2         | 0,45       |
| 73201    | R&D on social science   | 79,9         | 0,44       |
| 73202    | R&D on humanities   | 80,1         | 0,27       |
| 74150    | Management activities of holding companies                        | 34,9         | 0,22       |
| 67110    | Administration of financial markets                               | 48,6         | 0,13       |
| 65110    | Central banking   | 54,0         | 0,11       |
| 66020    | Pension funding   | 40,6         | 0,09       |
| 73105    | Interdisciplinary R&D on natural science and engineering          | 69,9         | 0,08       |
| 65210    | Financial leasing   | 31,2         | 0,06       |
| 73201    | Interdisciplinary R&D on humanities and social science            | 77,8         | 0,04       |
| 70110    | Development of selling of real estate                             | 40,5         | 0,02       |

Table A.1: Knowledge intense producer services with more than 30% knowledge intensity in 2007

| Innovation variable                                  | PATENT        | PANEL        | CIS P.        | ANEL         |
|--|---------------|--------------|---------------|--------------|
|  | Short-run     | Long-run     | Short-run     | Long-run     |
| IK <sub>11</sub> <sup>a</sup>                        | 0.000         | 0.000        | 0.000         | 0.000        |
|  | 0.016         | 0.01         | 0.009         | 0.01         |
| $IK_{12}$  | (0.033)       | (0.02)       | (0.106)       | (0.09)       |
|  | -0.004        | -0.00        | 0.025         | 0.02         |
| IK <sub>13</sub>                                     | (0.032)       | (0.02)       | (0.106)       | (0.09)       |
| IK <sub>21</sub>                                     | -0.001        | -0.00        | 0.059         | 0.05         |
|  | (0.111)       | (0.09)       | (0.190)       | (0.16)       |
| IK <sub>22</sub>                                     | -0.106        | -0.08        | 0.194         | 0.16         |
|  | (0.171)       | (0.13)       | (0.143)       | (0.12)       |
| IK <sub>23</sub>                                     | -0.103        | -0.08        | 0.038         | 0.03         |
|  | (0.168)       | (0.13)       | (0.150)       | (0.12)       |
| IK <sub>31</sub>                                     | -0.186        | -0.14        | -0.015        | -0.01        |
|  | (0.252)       | (0.19)       | (0.184)       | (0.15)       |
| IK <sub>32</sub>                                     | $0.182^{***}$ | $0.14^{***}$ | 0.235*        | $0.19^{*}$   |
|  | (0.047)       | (0.04)       | (0.132)       | (0.11)       |
| IK <sub>33</sub>                                     | $0.687^{***}$ | $0.54^{***}$ | 0.304 * *     | 0.25**       |
|  | (0.196)       | (0.16)       | (0.130)       | (0.11)       |
| Log Firm size, growth                                | $0.242^{***}$ | 0.27***      | 0.032         | 0.07         |
|  | (0.061)       | (0.05)       | (0.117)       | (0.10)       |
| Log TFP growth $t-1$                                 | -0.270***     |              | -0.209***     |              |
|  | (0.073)       |              | (0.027)       |              |
| $\log \mathrm{TFP}_{t-1}$                            | -0.226***     |              | -0.339***     |              |
|  | (0.057)       |              | (0.065)       |              |
| Domestic Uninational                                 | 0.039 * * *   | $0.03^{***}$ | 0.013         | 0.01         |
|  | (0.013)       | (0.01)       | (0.013)       | (0.01)       |
| Domestic multinational                               | $0.103^{***}$ | 0.08***      | $0.101^{***}$ | $0.08^{***}$ |
|  | (0.026)       | (0.02)       | (0.035)       | (0.03)       |
| Foreign owned multinational                          | $0.116^{***}$ | $0.09^{***}$ | $0.119^{***}$ | $0.10^{***}$ |
|  | (0.029)       | (0.02)       | (0.037)       | (0.03)       |
| Observations   | 196,027       |              | 20,076        |              |
| Unique Firms   | $31,\!208$    |              | $2,\!634$     |              |
| Laglimits  | $(3\ 1)$      |              | (2 1)         |              |
| Instruments  | 193           |              | 217           |              |
| AR(2)  | 0.106         |              | 0.894         |              |
| Hansen Overid.                                       | 0.831         |              | 0.779         |              |
| Difference-in-Hansen test for level equation         | 0.652         |              | 0.509         |              |
| Difference-in-Hansen test for lag dependent variable | 0.389         |              | 0.596         |              |

Table A.2: Dependent variable: TFP growth. Two-step system GMM Estimates. The IKvariable is treated as endogenous

Note: \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%

(a) Reference group (b) Reference group is domestic non-affiliated firms

Robust (GMM) standard error in parentheses. Year and sector dummies included

[IK11: Non R&I and Low access]; [IK12: Non R&I and Medium access]; [IK13: Non R&I and High access];

 $[\mathrm{IK}_{21}: \ \mathrm{Occ.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{Low} \ \mathrm{access}]; \ [\mathrm{IK}_{22}: \ \mathrm{Occ.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{Medium} \ \mathrm{access}]; \ [\mathrm{IK}_{23}: \ \mathrm{Occ.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{High} \ \mathrm{access}]; \ [\mathrm{IK}_{31}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{Low} \ \mathrm{access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{Medium} \ \mathrm{access}]; \ [\mathrm{IK}_{33}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{and} \ \mathrm{High} \ \mathrm{access}]; \ [\mathrm{IK}_{31}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{access}]; \ [\mathrm{IK}_{31}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{access}]; \ [\mathrm{IK}_{31}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{32}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ [\mathrm{IK}_{33}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ \mathrm{Access}]; \ [\mathrm{IK}_{33}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{And} \ \mathrm{High} \ \mathrm{Access}]; \ \mathrm{Access}]; \ [\mathrm{IK}_{33}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{Access}]; \ \mathrm{Access}]; \ [\mathrm{IK}_{33}: \ \mathrm{Pers.} \ \mathrm{R\&I} \ \mathrm{Access}]; \ \mathrm{Ac$ 

| Innovation variable           | TPF §         | growth        |
|-------------------------------|---------------|---------------|
|                               | PATENT        | CIS           |
| IK <sub>11</sub> <sup>a</sup> | 0.000         | 0.000         |
| $IK_{12}$                     | -0.004**      | -0.006        |
|                               | (0.002)       | (0.007)       |
| IK <sub>13</sub>              | 0.004*        | 0.004         |
|                               | (0.002)       | (0.008)       |
| IK <sub>21</sub>              | $0.016^{***}$ | -0.001        |
|                               | (0.006)       | (0.007)       |
| $IK_{22}$                     | 0.011         | 0.005         |
|                               | (0.008)       | (0.008)       |
| IK <sub>23</sub>              | 0.042***      | 0.011         |
|                               | (0.009)       | (0.009)       |
| IK <sub>31</sub>              | $0.034^{***}$ | 0.015         |
|                               | (0.010)       | (0.011)       |
| IK <sub>32</sub>              | $0.073^{***}$ | 0.040***      |
|                               | (0.012)       | (0.012)       |
| IK <sub>33</sub>              | 0.155 * * *   | $0.075^{***}$ |
|                               | (0.019)       | (0.013)       |
| Log Firm size, growth         | $0.310^{***}$ | 0.191***      |
|                               | (0.007)       | (0.020)       |
| Log TFP growth $_{t-1}$       | -0.330***     | -0.343***     |
|                               | (0.006)       | (0.020)       |
| $\log \mathrm{TFP}_{t-1}$     | -0.125*       | -0.134***     |
|                               | (0.068)       | (0.007)       |
| Domestic Uninational          | 0.015***      | -0.009*       |
|                               | (0.002)       | (0.005)       |
| Domestic multinational        | 0.044 * * *   | $0.031^{***}$ |
|                               | (0.003)       | (0.007)       |
| Foreign owned multinational   | 0.055***      | $0.036^{***}$ |
|                               | (0.003)       | (0.008)       |
| Observations                  | $196,\!027$   | 20,076        |

 Table A.3: Regression results. Dependent variables: value added and TFP growth. Pooled

 OLS estimates.

Note: \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%

(a) Reference group. (b) Reference group is domestic non-affiliated firms

Robust standard error in parentheses. Year and sector dummies included

[IK<sub>11</sub>: Non R&I and Low access]; [IK<sub>12</sub>: Non R&I and Medium access]; [IK<sub>13</sub>: Non R&I and High access]; [IK<sub>21</sub>: Occ. R&I and Low access]; [IK<sub>22</sub>: Occ. R&I and Medium access]; [IK<sub>23</sub>: Occ. R&I and High access]

[IK31: Pers. R&I and Low access]; [IK32: Pers. R&I and Medium access]; [IK33: Pers. R&I and High access]